Author's response to reviews

Title: Prospective cohort study of the relationship between neuro-cognition, social cognition and violence in forensic patients with schizophrenia and schizoaffective disorder

Authors:

Ken O'Reilly (oreillk5@tcd.ie)
Gary Donohoe (DONOGHUG@imx1.tcd.ie)
Ciaran Coyle (ciaran.coyle@hse.ie)
Danny O'Sullivan (danny.osullivan2@hse.ie)
Arann Rowe (arann.rowe@kcl.ac.uk)
Mairead Losty (mlosty@tcd.ie)
Tracey McDonagh (mcdonagt@tcd.ie)
Lasairiona McGuinness (lasairiona_mcguinness@hotmail.com)
Yvette Ennis (yvettetennis@eircom.net)
Elizabeth Watts (elizabeth.watts2@hse.ie)
Louise Brennan (louisepbrennan@eircom.net)
Elizabeth Owens (owensliz@hotmail.com)
Mary Davoren (davorem@tcd.ie)
Ronan Mullaney (rmullan@tcd.ie)
Mary Davoren (davorem@tcd.ie)
Zareena Abidin (zareenaabidin@yahoo.ie)
Harry G Kennedy (kennedh@tcd.ie)

Version: 7 Date: 27 June 2015

Author's response to reviews:

please see attached document